Author’s response to reviews

Title: A systematic review of patient access to medical records in the acute setting: practicalities, perspectives and ethical consequences

Authors:

Stephanie D'Costa (snd35@cam.ac.uk)

Isla Kuhn (ilk21@cam.ac.uk)

Zoe Fritz (zbmf2@cam.ac.uk; zoe Fritz@gmail.com)

Version: 2 Date: 06 Feb 2020

Author’s response to reviews:

We thank Reviewer two for their editorial suggestions; we have accepted them all and made appropriate revisions. We have also changed and tightened up the abstract.